Author’s response to reviews

Title: RePAIR Consensus Guidelines  Responsibilities of Publishers, Agencies, Institutions, and Researchers in Protecting the Integrity of the Research Record

Authors:

Carolyn Broccardo (cbroccar@uwyo.edu; drmama@comcast.net)
Noemie Aubert Bonn (noemie.aubertbonn@uhasselt.be)
Laura Bandura-Morgan (laura.bandura-morgan@ncn.gov.pl)
Jeffrey Beall (Jeffrey.beall@ucdenver.edu)
Catherine Bens (cat.bens@colostate.edu)
Monica Bradford (mbradfor@aaas.org)
Lyndon Branfield (Lyndon.Branfield@springernature.com)
Arthur Buchberg (arthur.buchberg@gmail.com)
Ferric Fang (fcfang@u.washington.edu)
Anna Keith (agarber29@gmail.com)
Barb Houser (barb.houser@colostate.edu)
Jennifer Nyborg (jknyborg@gmail.com)
Kelly Perry (kellyella@comcast.net)
Charon Pierson (charon.pierson@gmail.com)
Kenneth Pimple (ken@teachrcr.us)
Kaoru Sakabe (ksakabe@asbmb.org)
Dan Wainstock (daniel_wainstock@hms.harvard.edu)
Tamara Welschot (tamara.welschot@springer.com)
Brad Woods (brad.woods@colostate.edu)
Alice Young (alice.young@ttu.edu)
Version: 1 Date: 21 Sep 2018

Author’s response to reviews:

I have made a few changes in the manuscript to address some comments in the review by Zoe Hammatt. I also updated the email address for Mr. Branfield.